Author’s response to reviews

Title: Anti-Methicillin-resistance Staphylococcus aureus (MRSA) compounds from Bauhinia kockiana Korth. and their mechanism of antibacterial activity

Authors:
Joo Kheng Goh (goh.joo.kheng@monash.edu)
Yik Ling Chew (chewyl@ucsiuniversity.edu.my)
Adlina Maisarah Mahadi (ammah5@student.monash.edu)
Kak Ming Wong (wongkakming@hotmail.com)

Version: 3 Date: 06 Feb 2018

Author’s response to reviews:

Responses from authors

Please make the following revisions to your manuscript:

1. Please include the email addresses of all authors on the title page of your manuscript.
Response: All authors email addresses and affiliations are included on the title page of the manuscript.

2. Please separate the Results and Discussion section, in order to adhere to journal submission requirements.
Response: Result and Discussion sections are separated into individual sections.

3. Please add an Ethics section to your Declarations and under it please write, "not applicable".
Response: ‘Ethics approval and consent to participate’ is included under Declarations.

4. Please add a Funding section to your Declarations.

Response: ‘Funding’ is included under Declarations.